How to Start the Netscape Browser at M.I.K

MA 113 – 001, 002, 003

Wednesday, August 28, 1996

Class materials for this course will be handed out in class and made available on the internet. In order to access the network copy, you will need to use a world-wide-web browser program, like Netscape. To use it, go to any of the pc labs on campus, e.g. there is one on the second floor of Margaret King Library – South. There is a different procedure depending on whether or not the machine is a personal computer or a Macintosh.

If you have problems, just ask one of the monitors in the room for help. From a PC:

1. Click the left mouse button on the “Start” icon, located in the bottom left corner of the screen.

2. Move the mosue cursor up over “Programs”. Click the left mouse button.

3. Move the mouse cursor over “Internet Applications”. Click the left mouse button.

4. Move the mouse cursor over “Netscape Navigator”. Click the left mosue button.

5. Netscape should start and display on the screen. To access the class materials, type the following into the “Location” entry box and hit the Enter key: http://www.ms.uky.edu/~ken

6. When you are done, you can quit Netscape by pulling down the “File” menu, and clicking on “Exit”.

From a Mac:

1. Double click the mouse button on the “Mac Drive” icon located in the upper right side of the screen.

2. Double click the mouse button on the “Communications and Information” folder.

3. Double click the mouse button on the “Netscape Navigator 2.02” folder. 4. You can access the class materials by typing the following into the “Location” entry box and hitting the Enter key: http://www.ms.uky.edu/~ken